Daily Report for 15 June 2010, Ocean Veritas.

The vessel arrived at the first station (OV047), 5Km WSW of the well head at 0720h. The rationale for this choice of station was that the RV Brooks McCall had detected a small but significant degree of toxicity in a sub-surface plume at this station (as BM 077) during that vessel’s previous cruise, as determined by the Rototox test. Although no plume was evident at this station today, two samples were collected for toxicity bioassay.  As on the previous day, at all stations visited we also deployed a 38KHz Acoustic Doppler Current Profiler (ADCP), off the starboard side, at a depth of approximately 3m, and 7m out from the starboard side.

The second station occupied (OV048) was due SW from the well head, appx. 1.5Km from the source. As such the station has assumed the role of a ‘reference station’, having been visited several times in May and June by both the RV Brooks McCall and RV Ocean Veritas (e.g. as OV011 on May 29th). No significant fluorescence was detected at this station. This result was in remarkable contrast to several earlier records from this station most notably on May 29th, when the largest fluorescence yet recorded by either vessel was found at this site. This reinforces the recent observations by both vessels of negligible fluorescence signals at this site, indicating that the dispersed plume has moved away from the SW quadrant. 
In line with most recent observations that the plume has shifted ‘clockwise’ from SW → NW → NE we established the third station of the day (OV049) 5Km NNW of the well head. Again the trace showed no significant fluorescence. This result indicated that the plume had moved from the position in the NW quadrant that it occupied during the last cruise by the RV Brooks McCall. That vessel had tracked a deep fluorescence signal to a point due N of the source, and our observations from yesterday indicated that the plume had moved into the NE quadrant.
 For station 4 (OV050) we moved to a position 5Km NNE of the source; 2Km further out from yesterday’s station OV043.  A small, weak fluorescence signal was recorded at this station at 1070m. 
For the fifth and final station of the day (OV051) we followed the NNE transect back towards the well head, 2Km out from the source. This was the station aborted yesterday due to high benzene levels. Moderately high benzene levels, around 1ppm were recorded at this station, although well below the 9+ ppm levels recorded yesterday. Sampling at this station was completed using respirators. The fluorescence trace at this station was characterized by weak peaks at 1070-1100m and 1100-1130m, weak ‘spikes’ at 350m and 820m, and a moderate spike at 800m.

Overall, the fluorescence data indicate little evidence of a substantial sub-surface dispersed oil plume at points west in the vicinity of the well head. Evidence from both vessels indicates and seems to have shrunk, and moved into the NE quadrant. We will continue to track the plume to the north and east of the well head tomorrow.
Delays in hatching of rotifers necessitated postponing the set-up of four samples from yesterday for bioassay. These were from station OV046 at 3m, 500m, 1120m and 1170m. Today two samples were selected for bioassay from today’s first station (OV047), as a follow-up to the small but significant toxic ‘hit’ recorded by the RV Brooks McCall from station BM77 last week. Although no fluorescence was detected in our samples at a depth where a plume might be expected, we selected water from OV047 at 1012m and 146m for testing. Four ‘treatments’ (depths) were also selected for bioassay from the only station sampled today that showed any significant evidence of a dispersed oil plume (OV051). The depths, 3m, 400m, 800, and 1080m were selected on the basis of fluorescent trace characteristics. Exposures for all these bioassays are still in progress at the time of this report.  Although data from these bioassays will not be available for at least another 24h, the spread sheet showing sample ID and treatments is included in today’s attached files.
Details of stations visited, CTD data, fluorescence profiles and a summary of LISST particle analyses are also included in files submitted today. The ADCP was as again deployed at all stations visited today, although no data are reported at this time.
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